Author's response to reviews

Title: Family physicians' effort to stay in charge of the medical treatment when patients have home care by district nurses. A grounded theory study

Authors:

Sonja Modin (sonja.modin@sll.se)
Lena Törnkvist (lena.tornkvist@sll.se)
Ann-Karin Furhoff (anna-karin.furhoff@ki.se)
Ingrid Hylander (ingham@ibv.liu.se)

Version: 5 Date: 9 June 2009

Author's response to reviews: see over
Title: A grounded theory study of family physicians’ effort to stay in charge of the medical treatment for patients with home care by district nurses.

Authors:
Sonja AE Modin (sonja.modin@sll.se)
Lena Törnkvist (lena.tornkvist@sll.se)
Ann-Karin Furhoff (anna-karin.furhoff@ki.se)
Ingrid Hylander (inghy@ibv.liu.se)

Version: 4 Date: 9 June 2009

Author's response to requested revisions:
Major revisions (We require the author to make these changes)

Reference 13, please provide full and accurate details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city.
We have had trouble to find the adequate Reference type in EndNote. This a document found on the home page of WONCA EUROPE (The European Society of General Practice/ Family Medicine) the European branch of WONCA. Described as: “Wonca is an acronym for the World Organization of National Colleges, Academies and Academic Associations of General Practitioners/Family Physicians, or World Organization of Family Doctors for short”. The document which includes 31 pages is a definition of Family medicine/General practice formulated for this organisation in 2002 and revised in 2005. We have now used the Report format as that seemed to be the most adequate and as Series title we have entered the webpage where you can find the document.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Tracking - Please remove all tracking from the manuscript.
We have tried to remove all tracking from the document

Tables - Please do not include vertical lines in your tables.
We have taken away all lines in the tables

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:
We have gone through the figures and made sure they cropped as closely as possible